Author's response to reviews

Title: The relationship between proton pump inhibitor use and serum magnesium concentration among hemodialysis patients: a cross-sectional study

Authors:

Paraish S Misra (Paraish.misra@mail.mcgill.ca)
Ahsan Alam (ahsan.alam@mcgill.ca)
Mark L Lipman (mark.lipman@mcgill.ca)
Sharon J Nessim (sharon.nessim@mcgill.ca)

Version: 5
Date: 31 July 2015

Author's response to reviews:

As requested, we have revised the manuscript to adhere to the editorial requests, including addition of the STROBE checklist, ensuring that the abstract in the BMC system matches the one in the manuscript, adding an acknowledgements section and moving the reference section ahead of the tables.